Author's response to reviews

Title: The positive association between number of children and obesity in Iranian women and men: Results from the National Health Survey

Authors:

Enayatollah Bakhshi (Bakhshi@razi.tums.ac.ir)
Mohammad Reza Eshraghian (eshraghian@yahoo.com)
Kazem Mohammad (mohamadk@tums.ac.ir)
Abbas Rahimi Foroughshani (rahimifo@tums.ac.ir)
Hojat Zeraati (zeraatih@tums.ac.ir)
Akbar Fotouhi (afotouhi@tums.ac.ir)
Fraidon Siassi (siassif@sph.tums.ac.ir)
Behjat Seifi (behins@yahoo.com)

Version: 7 Date: 10 May 2008

Author's response to reviews:

Dear Natali Pafitis MSc,
Assistant Editor
BMC-Series Journals

According to your final comment, we contacted our university’s Ethics Committee and have added some statements to the section of "Data set examined" and lines 9-16 of this section now appears as follow:

Data from the National Health Survey were considered in this investigation. These data were collected by the National Research Center of Medical Sciences and, although are not openly available, are presented partially at the Department of Biostatistics and Epidemiology/Tehran University of Medical Sciences for research. We focused our study on the respondents who lived in Tehran Province only, and the analyzed data included 2728 women and men (1364 couples) aged 20-75 years. We excluded pregnant women from the analyses. This study has been approved by Ethics Committee of Tehran University of Medical Sciences.

Best Wishes

Corresponding Author
Dr. M. R. Eshraghian